Author's response to reviews

Title: An adaptogenic role for omega-3 fatty acids in stress; a randomised placebo controlled double blind intervention study (pilot)

Authors:

   Joanne Bradbury (jbradbury@byrononline.net)
   Stephen P Myers (smyers@scu.edu.au)
   Chris Oliver (coliver@scu.edu.au)

Version: 2 Date: 7 October 2004

Author's response to reviews:

Dear Reviewers,

Thank you for your comments and directions. I will raise the issue of ISRCTN registration with the co-authors at a meeting next week, with the view of registering all our future trials.

I have made the requested formatting changes. These include:
Abstract - first sentence has been changed from 'There evidence...' to read 'There is evidence...'
Abstract - was moved to the second page
Qualifications - 'Professor' was deleted from the authors qualifications
Affiliations - the addresses of the author's affiliations has been given in full
Corresponding author - has been clearly indicated on the title page
Tables - have been placed below their legend after the references on the manuscript
Figures - have been cropped to minimise the white space surrounding the image
Figure legend - has been removed from the image.

Yours faithfully,
Joanne Bradbury